Author’s response to reviews

Title: Computed Tomography Volumetry of esophageal cancer - The role of Semiautomatic assessment

Authors:
Yi-Hua Zhang (yi-hua.zhang@ki.se)
Michael Fischer (dr.michael.fischer@gmail.com)
Henrik Lehmann (henrik.lehmann@karolinska.se)
Åse Johnsson (ase.johnsson@vgregion.se)
Ioannis Rouvelas (Ioannis.Rouvelas@ki.se)
Gunnar Herlin (gunnar.herlin@karolinska.se)
Lars Lundell (lars.lundell@ki.se)
Torkel Brismar (torkel. brismar@gmail.com)

Version: 4 Date: 23 Jan 2019

Author’s response to reviews:

1. Identifying information

Currently, the age and gender information present in Table 1, in combination with other identifiers, may compromise patient/participant anonymity.

If you do not have consent to publish, please amend the table to address this issue, by removing this information or by providing ages as age-ranges.

Information potentially compromising patient anonymity has been removed from Table 1 and summarized in text.

2. Clean manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional-supplementary files are cited within the text.
A version with no tracked changed has been uploaded.